Author's response to reviews

Title: Protective measures and H5N1-seroprevalence among personnel tasked with bird collection during an outbreak of avian influenza A/H5N1 in wild birds, Ruegen, Germany, 2006

Authors:

Wei Cai (caiw@rki.de)
Brunhilde Schweiger (schweigerb@rki.de)
Udo Buchholz (buchholzu@rki.de)
Silke Buda (budas@rki.de)
Martina Littmann (martina.littmann@lagus.mv-regierung.de)
Jörg Heusler (joerg.heusler@landkreis-ruegen.de)
Walter Haas (haasw@rki.de)

Version: 7 Date: 13 October 2009

Author's response to reviews:

Dear Editorial Production Team,

We have checked the additional material (questionnaire_english) again to the best of our knowledge for typing errors and other errors.

With best regards
Walter Haas

Priv. Doz. Dr. med. Walter Haas
Head of the Unit for Respiratory Infections Department of Infectious Disease Epidemiology Robert Koch Institute DGZ-Ring 1
13086 Berlin (Weißensee)
Germany

Pho: ++49 30 18754-3431
Fax: ++49 30 18754-3533

***The Robert Koch Institute is a Federal Institute within the portfolio of the Federal Ministry of Health
GERMANY***